Soil B activity

16:00 - 17:00: Steve O. noticed issues with the TPO1 at soil B. This problem was in addition to the Tsoil going out today. So we decided to replace the

mote. Unfortunately the spare is being used for soil.a2. An attempt was made to get a mode Steve O. brought with him. He was going to play with it
working'

on the chirpsonic software. The attempt failed; however, we did get the Tsoil going again. As noted earlier water was found in the bottom of the mote box.
The Tsoil

problem may have been related to the water which has now dried out. We will monitor.
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